# The Influence of Digital Technologies on Knowledge Management in Engineering: A Systematic Literature Review

Yuxin Yao, Eann A. Patterson, Richard J. Taylor

**Abstract** — Digital technologies are gaining widespread acceptance in engineering and offer opportunities for collating and curating knowledge during and beyond the life cycle of engineering products. Knowledge is central to strategy and operations in most engineering organizations and digital technologies have been employed in attempts to improve current knowledge management practices. A systematic literature review was undertaken to address the question: how do digital technologies influence knowledge management in the engineering sector? Twenty-seven primary studies were identified from 3097 papers on these topics within the engineering literature published between 2010 and 2022. Four knowledge management processes supported by digital technologies were recognized: knowledge creation, storage and retrieval, sharing and application. In supporting knowledge management, digital technologies were found to have been acting in five roles: repositories, transactive memory systems, communication spaces, boundary objects and non-human actors. However, the ability of digital technologies to perform these roles simultaneously had not been considered and similarly knowledge management had not been addressed as a holistic process. Hence, it was concluded that a holistic approach to knowledge management combined with the deployment of digital technologies in multiple roles simultaneously would likely yield significant competitive advantage and organizational value for organizations in the engineering sector.

Index Terms— Engineering, information technology and systems, knowledge and data engineering tools and techniques, knowledge management, social technologies

### **1** INTRODUCTION

KNOWLEDGE is central to strategy and operations in most engineering organizations [1]. In the past twenty years, knowledge has been viewed as one of the most strategically significant organizational resources and a source of competitive advantage [2, 3]. Alongside this, the central role of knowledge as the essence of many organizations has led to an increasing interest in knowledge management [4].

Knowledge management is defined as the process of identifying, capturing, sharing, and utilizing the collective knowledge in an organization to help the organization compete [5]. In highly knowledge intensive industries such as engineering, efficient knowledge management is recognized as the key to success [3].

Digital entities, such as digital twins in the aerospace industry [6] and digital environments in the nuclear power industry [7], are transforming engineering processes throughout the life cycle of products and have the potential to act as persistent sources of information about products for generations of stakeholders. Simultaneously, advances in computer-based technologies have the potential to systematize, enhance and expedite knowledge management

[5, 8]. However, organizations sometimes promote changes in technology without considering the need for a commensurate alignment of people, structure, processes and culture [9]. In these situations, it is only after the technology capabilities have been put in place, that many organizations realise that technology on its own is not enough to address knowledge management problems [10]. "People, process, technology" is a popular term in knowledge management studies. Bhatt [11] acknowledged that these three elements collectively influence the effectiveness of knowledge management. Technology is only as good as the processes that wrap around it, while the processes are only as effective as the people that use them [12]. Hence, in the light of the radical changes in engineering practice emerging through the use of digital entities throughout a product's lifecycle, the objective of this study was to conduct a systematic literature review of the relationships between these digital technologies and knowledge management in the engineering sector.

A growing number of review studies have been undertaken which analyze digitally supported knowledge management in knowledge-intensive sectors. For example, Nidhra et al. [13] investigated knowledge transfer challenges and mitigation strategies in global software development; Ahmed et al. [14] studied the current state of research regarding social media used for knowledge sharing; Inkinen [15] reviewed the empirical research on knowledge management practices and firm performance; and Manesh et al. [16] studied the knowledge management changes and future avenues in the fourth industrial revolution, in which digital technology is increasing interconnectivity and

<sup>•</sup> Y. Yao is with the School of Engineering, University of Manchester, Email: yuxin.yao@manchester.ac.uk.

E.A. Patterson is with the Department of Mechanical, Materials and Aerospace Engineering, University of Liverpool. E-mail: eannp@liverpool.ac.uk.

<sup>•</sup> R.J. Taylor is with the Department of Mechanical, Aerospace and Civil Engineering, University of Manchester. E-mail: richard.taylor-4@manchester.ac.uk.

automation producing rapid changes in both engineering industry and society. However, among these reviews, few have focused on social factors related to knowledge management and almost none conducted their review within engineering. Thus, the aim of this review is to close this gap by considering the social factors related to knowledge management in the context of the engineering sector. Specifically, these gaps are highlighted by interrogating the literature for research outputs associated with the use of digital technology, associated with Industry 4.0, in knowledge management processes in all engineering sectors. The toplevel research question established as a basis for this review is: how do digital technologies influence knowledge management in the engineering sector? The answer to this question, derived from this review, is that digital technologies are having some influence on knowledge management in the engineering sector; however, only on individual or isolated processes rather than in a holistic manner, and with little or no impact on social practices.

This review is structured with six sections. Section 2 provides a background of knowledge and knowledge management within engineering. Section 3 describes the methodology through which this review was conducted. Section 4 provides a summary of the content of the papers selected in the systematic review. Section 5 discusses the findings related to the research question above and conclusions are presented in Section 6.

### 2 BACKGROUND

### 2.1 Knowledge

Defining knowledge has occupied the minds of philosophers and researchers since the classical Greek era and has led to various epistemological debates [5]. This study does not discuss the diversity of knowledge definitions. However, it is necessary to make a clear distinction between information and knowledge. Information is defined as structured and understandable data, organized in order to be a useful input to knowledge [17]. Knowledge has many definitions in the literature, each of which is applicable in certain circumstances. One definition provided by Davenport and Prusak [18] is: "Knowledge is information with the most value and is consequently the hardest form to manage. It is valuable precisely because somebody has reflected on the knowledge, added their own wisdom to it, and considered its larger implications." According to Fahey and Prusak [19], knowledge does not exist independently of a knower: it is shaped by one's needs as well as one's initial stock of knowledge. Thus, managing information is by nature different from managing knowledge. The key factor that distinguishes knowledge from information is "people". As Ruggles [10] said, "if the people issues do not arise, the effort underway is probably not knowledge management. If technology solves your problem, yours was not a knowledge problem."

### 2.2 Knowledge Management in Engineering

"Engineering" is defined by the Oxford English Dictionary as "the branch of science and technology concerned with the development and modifications of engines, machines, structures or other complicated systems and processes using specialized knowledge or skills." [20]. This definition carries the implication that managing specialized knowledge and skills is basic to engineering organizations. Madhavan [21] described engineers as integrators who pull ideas together from multiple streams of knowledge. This indicates that engineers need to work within a wide spectrum of knowledge rather than single strands, which means it is critical to facilitate interactions and connections between entities involved in engineering undertakings.

However, due to the large division of labor across geographical boundaries, between multi-functional business units, multi-tier supply chains and multi-disciplinary projects, the preservation, transfer and inheritance of knowledge presents many challenges [22, 23]. In dispersed organizations, employees often cannot identify what is known by other colleagues as it remains hidden in knowledge silos. Creating a sustainable knowledge sharing culture is therefore a highly complex task [24]. Moreover, breakdowns of communication are often experienced due to the inevitability that involved stakeholders sometimes have various interests and professional backgrounds with different expertise, terminologies, and perspectives [25].

# 3 METHOD

A systematic literature review is a means of identifying, evaluating and interpreting all available research relevant to a particular research question, topic area, or phenomenon of interest [26] and is known as a tertiary review when it considers prior systematic reviews. A systematic review differs from traditional narrative reviews in employing a replicable, scientific and transparent process which generates an audit trail of the reviewer's decisions, procedures and conclusions [27]. While a systematic review provides some quantitative information about the available literature, it does not yield detailed statistical data about the relationships between authors and publications provided by a bibliometric review or mapping. In this study, a systematic review was performed because the research relevant to the particular topic area, defined by a research question, was of interest. The guidelines for systematic reviews proposed by Kitchenham and Charters [26] and Tranfield et al. [28] were implemented using the recommended three phases: planning, conducting and reporting, which are explained as follows:

#### 3.1 Planning

A comprehensive review protocol was defined to guide the study and provide a clear plan for the review's progress. An initial scoping review was first conducted to confirm the need for this systematic literature review. Three main keywords for the search were selected based on the research question and a search was conducted in Web of Science for journal articles only published in the period from 2010 to 2022 in the English language. This search revealed a scarcity of recent review papers on this topic, only 13 reviews were identified. The search string employed was: ("knowledge management" AND "social\*" AND "digital\*"). Within these 13 reviews, most papers were traditional narrative literature reviews except for two systematic literature reviews. None of the identified reviews focused on digitally supported knowledge management in the engineering sector. Therefore, this confirmed the identified gap and the need to undertake this study.

Synonyms for the three main keywords were added to the search string. The final search string was formed in three parts: "knowledge management" and all the related processes; "social" and its synonyms; "digital" and its synonyms. These three parts were joined with an "AND" operator, and within each part, the synonyms were joined with an "OR" operator. The search string was employed to search the title, abstract and keywords of journal articles available in English and published from 2010 to 2022. The full search string used is shown in Table 1.

Five databases, namely: ACM Digital Library, IEEE explore, ProQuest, Scopus, Web of Science, were selected as the main sources for the research. These databases were selected because they cover a broad range of subject fields and were available to the authors free of charge. All the search results were organized using the EndNote software.

The use of multiple databases with overlapping scopes combined with the inclusion of multiple synonyms for the three main keywords ensured that the probability of omitting a relevant study was low, although it did increase the probability of a large number of duplicates, however this was an acceptable cost to ensure the completeness of the review.

TABLE 1 Search String

Keyword	Synonyms				
Knowledge	"knowledge sharing" OR "knowledge transfer"				
management	OR "tacit knowledge" OR "knowledge flow"				
	OR "knowledge capture" OR "knowledge				
	creation" OR "knowledge application" OR				
	"knowledge acquisition" OR "knowledge				
	retention" OR "knowledge exchange"				
Digital*	"cyber*" OR "virtual*" OR "digitiz*" OR				
	"digitis*"				
Social*	"interact*" OR "communicat*" OR "inter-*" OR				
	"intra-*" OR "collaborat*"				
Complete search	("knowledge management" OR "knowledge				
string	sharing" OR "knowledge transfer" OR "tacit				
	knowledge" OR "knowledge flow" OR				
	"knowledge capture" OR "knowledge creation"				
	OR "knowledge application" OR "knowledge				
	acquisition" OR "knowledge retention" OR				
	"knowledge exchange") AND ("digital*" OR				
	"cyber*" OR "virtual*" OR "digitiz*" OR				
	"digitis*") AND ("social*" OR "interact*" OR				
	"communicat*" OR "inter-*" OR "intra-*" OR				
	"collaborat*")				

### 3.2 Conducting

After completing the search process and removing 733 duplicates, 3097 papers remained. The remaining papers were filtered using a set of inclusion and exclusion criteria. The aim of applying inclusion and exclusion criteria was to ensure that the primary studies selected for review were

relevant to the research question. The inclusion and exclusion criteria used in this review are shown in Table 2. The inclusion and exclusion criteria were applied to screen the remaining papers by reading the title, abstract and keywords of each paper.

The full text of papers that remained after applying the

TABLE 2	
---------	--

Inclusion Criteria	Exclusion Criteria	
Available in full text	Not available in full text	
Published in the time period	Outside the search timespan	
selected		
Available in English	Not available in English	
Related to the research questions	Not related to the research ques-	
	tions	
Published in selected digital	Duplicated studies	
databases	-	

exclusion and inclusion criteria was read and a quality assessment conducted by appraising the internal consistency and rigor of each study [29]. Based on the generic criteria proposed by Kitchenham and Charters [26], five specific quality assessment criteria were used in this review:

- 1. Is the topic addressed in the study related to knowledge management processes in a digital environment?
- 2. Are the knowledge management processes addressed in the study clearly specified?
- 3. Is the digital technology/framework considered by the study clearly specified?
- 4. Is the research methodology clearly specified?
- 5. Are the results of the studies relevant to this systematic review?

Following examples in the literature (e.g., [13, 30]), each criterion was scored from 0 (no compliance), 0.5 (partial compliance) to 1 (full compliance); hence, a study's highest possible score was 5, while its lowest possible score was 0. In this review, the quality of each paper was considered to be high if it scored greater than or equal to 3. A paper that scored between 2 to 3 was considered to be of medium quality. A paper that scored equal to or less than 2 was considered to be low quality and was excluded. No papers scored less than 2.5, and thus no papers were excluded as a result of this assessment. 27 papers were selected as primary studies (i.e., individual studies contributing to the systematic literature review) for further analysis. Fig. 1 shows the selection process of this review.

### 3.3 Reporting

A data extraction step identified and extracted relevant data from the primary studies. A data extraction form was developed to accurately record the information gathered through the review [26]. This process was conducted by reading the full text of each study and cataloguing the extracted information using Microsoft Excel spreadsheets and EndNote. The extracted elements including study ID (using "S" + "number" to represent each study, e.g., S1 represents [31]), publication year, research field, main knowledge management process(es), technology. The completed data extraction form is shown in Table 3.

TABLE 3 Data Extraction Form

S1	[28]	2010	Not specified	Knowledge creation	Social-tagging systems, pattern-based task- management systems, wikis. (Social media)
S2	[42]	2010	Manufacturing	Knowledge storage and retrieval	a Multi-Agent System on a Virtual Reality platform (Visualization technologies)
S3	[47]	2013	Construction	Knowledge sharing	Building Information Modelling (BIM)
S4	[32]	2013	Not specified	Knowledge storage and retrieval	Big data supported knowledge recorder tool
S5	[35]	2014	Financial	Knowledge sharing	Enterprise social media
S6	[36]	2015	Engineering design	Knowledge sharing	Social media
S7	[29]	2015	Coal mining	Knowledge storage and retrieval	A web-based interactive database (Content management system)
S8	[43]	2015	Nuclear	Knowledge sharing	3D simulation, virtual reality, advanced user interfaces, mobile and wearable devices, geographical information systems. (Visualisation technologies)
S9	[37]	2016	Aerospace	Knowledge sharing	Micro-blogging (Social media)
S10	[44]	2017	Engineering design	Knowledge storage and retrieval	Virtual Aided Design Engineering Review system (Visualisation technologies)
S11	[38]	2017	Multiple industries	Knowledge sharing	Social networking sites (Social media)
S12	[33]	2017	Oil and gas	Knowledge creation	Big data
S13	[21]	2017	Nuclear	Knowledge sharing Knowledge storage and retrieval	Multiple technologies
S14	[49]	2018	Manufacturing	Knowledge sharing	Augmented reality and sensor technology (Cyber physical system)
S15	[22]	2018	Aerospace	Knowledge sharing	Web 2.0: Blogs, Wikis, Forums, social networking sites, etc. (Social media)
S16	[55]	2018	Manufacturing	Knowledge creation Knowledge sharing	Augmented reality, Web 2.0, Data mining (Multiple technologies)
S17	[40]	2018	Multiple industries	Knowledge creation Knowledge sharing	Enterprise social networking system (Social media)
S18	[39]	2018	Manufacturing	Knowledge sharing	Enterprise social media
S19	[30]	2018	Manufacturing	Knowledge storage and retrieval	Content management systems
S20	[54]	2018	Manufacturing	Knowledge creation	Multiple technologies.
S21	[50]	2019	Manufacturing	Knowledge sharing	Cyber-physical systems: Augmented reality, sensor technology
S22	[51]	2019	Manufacturing	Knowledge creation Knowledge sharing	Industry 4.0. Service-oriented Digital Twin. (Cyber-physical systems)
S23	[41]	2020	Software development	Knowledge sharing	Social media
S24	[52]	2020	Manufacturing	Knowledge creation	Industry 4.0. Cyber-physical systems
S25	[23]	2020	Construction	Knowledge sharing	Immersive VR systems and a multitouch table (Visualisation technologies)
S26	[31]	2022	Railway	Knowledge storage and retrieval	Content management system
S27	[45]	2022	Manufacturing	Knowledge sharing	Visualisation technologies





rig. 5 Distribution of Frinary Studies Across Resea

Fig. 1 Study Selection Process

Fig. 2 shows the chronological distribution of the primary studies from 2010 to 2022 and their content in terms of four knowledge processes, namely, knowledge creation, knowledge storage and retrieval, knowledge sharing and knowledge application [5]. Knowledge sharing has been most frequently studied during the last ten years while knowledge creation has been the focus of a number of studies between 2017 to 2020.



Knowledge sharing & Knowledge storage and retrieval

Fig. 2 Chronological Distribution of Primary Studies Showing Frequency of Focus on Four Knowledge Processes

Fig. 3 shows the distribution of research fields described by the primary studies. Manufacturing (10 studies) is the area with the most publications.

# 4 CONTENT SUMMARY

This section provides a description of how different digital technologies feature within the twenty-seven primary studies. To facilitate this description, the studies have been grouped based on the technology that is their principal focus, into seven technology clusters: Content Management Systems (CMS); big data; social media; visualization technologies; Building Information Modelling (BIM); Cyber-Physical Systems (CPS), and multiple technologies.

### 4.1 Content Management

Two of the primary studies investigated how content management systems have supported knowledge management. Kirsch et al. [32] introduced an interactive database called RISKGATE to support knowledge sharing between multiple stakeholders in the coal mining industry. The RISKGATE, in the form of a digital web-based tool, supported knowledge management by capturing inter-organizational expert knowledge and facilitating dissemination of the knowledge to field practitioners.

Content management systems (CMS) have been widely used in non-engineering sectors as one of the most important information and communication technologies for managing organizational information and knowledge because it has good capabilities, flexibility and extendibility, but there are few applications of CMS for managing engineering knowledge [33]. Wan et al. [33] proposed a collaborative maintenance planning system based on advanced content management to connect all stakeholders involved in a machine tool's lifecycle. Machine tool manufacturers, users, service providers and parts suppliers would all acquire product knowledge, workflow, maintenance planning and lessons learnt through the same platform. The prototype system was evaluated using a machine tool maintenance case and they showed that the proposed system could potentially enhance the efficiency and effectiveness Abbas et al. [34] conducted a case study at a Dutch railway organization consisting of a group of test laboratories. The case study was based on a content management system, which the authors developed, that supported building collaboration by allowing participants in the laboratories to share knowledge. The system primarily focused on improving the findability and learnability of shared knowledge. They evaluated the performance of the developed system by comparing it to the existing system in terms of file search time and usability of the lessons learned function. They concluded that the proposed knowledge management system facilitated the building of digital collaborations.

### 4.2 Big Data

Two of the primary studies considered how big data has been applied for knowledge management. In order to study knowledge discovery from diverse data sources, Sukumar and Ferrell [35] introduced an enterprise data discovery and knowledge recorder tool called Schema Ex-ploration and Evolving Knowledge Entity Recorder (SEEKER). The SEEKER software acted as a virtual warehouse of institutional memory within and across enterprise data assets, a repository of the evolution of data elements, queries and schemas as function of time, and a visualization and learning tool for personnel new to the system.

Sumbal et al. [36] explored the relationship between big data and knowledge management by conducting a qualitative study using semi-structured interviews in the oil and gas sector. They found that actionable knowledge was generated through a combination of tacit knowledge of people and explicit knowledge obtained from big data to support decision making in the organization.

### 4.3 Social Media

Nine of the primary studies focused on using social media to facilitate knowledge management. Enterprise social media is defined as a set of enterprise-wide, internet-based technologies that allow users to easily create, edit, evaluate and link to content and to content creators [37]. The applications of enterprise social media include wikis, blogs, social tagging systems, microblogs, and enterprise social networking systems. Kimmerle et al. [31] described how collaborative knowledge building, aimed at producing new knowledge, occurs in an organization when people interact with each other supported by social media technologies, including social-tagging systems, pattern-based task management systems and wikis. Using micro-blogging as the mechanism, Beck et al. [38] proposed and empirically tested a multilevel model of the characteristics of knowledge seekers, knowledge contributors and their relationships using a dataset of 15,505 messages from microblogging between employees in an enterprise. Gopsill et al. [39] developed a social media tool to support engineering design communication. They went on to validate the requirements that underpin the social media framework and evaluate the impacts of the tool on engineering work, engineering artefacts and engineering project management. The tool was applied to an eleven-week Formula Student race car design project. Evans et al. [40] proposed a conceptual framework to illustrate how organizations in the aerospace industry may improve the capture of employee knowledge to address production development problems through the use of micro-blogging for crowdsourcing ideas and solutions.

Scuotto et al. [41] quantitively analyzed the relationships between social networking sites, innovation performance and absorptive capacity, which was defined as the ability to support enterprises to acquire knowledge from external environments. The partial least squares (PLS) regression results showed that social networking sites have a positive role in affecting both the capacity to absorb knowledge and the innovation performance of small and mediumsized enterprises. Evans et al. [24] used a qualitative research method for a five-year participant-observation study. They developed a framework for virtual knowledge sharing and collaboration in the dispersed development of aerospace products based on a set of social software platforms. The results indicated that social software technologies offered a more openly innovative environment where people shared knowledge more easily and effectively across geographical and functional boundaries. Van Osch and Steinfeld [42] indicated that the visibility of resources on enterprise social media (ESM) could be leveraged strategically to evoke diverse network structures and different boundaryspanning activities across working groups within the organizations.

Qi and Chau [43] also used PLS to investigate the relationship between enterprise social network system (ESNS) usage, organizational learning and knowledge management processes. They found that ESNS usage was an important antecedent of knowledge creation and knowledge sharing, and an important contributor to organizational learning. They also found that the exchange between the organization's social system and an individual's cognitive system supported the development of new knowledge.

Ali et al. [44] applied PLS to analyze the behavior of sixtyone software development teams and tested the influence of social media on absorptive capacity, team innovation performance, and transactive memory, which was described as a socio-cognition system that helps team members benefit from distributed knowledge resources. They recommended that software development teams should consider adopting social media that facilitates the development of capabilities to acquire and integrate knowledge from internal and external resources in order to enhance innovation.

# 4.4 Visualization Technologies (Virtual Reality and Augmented Reality)

Five of the primary studies examined how visualization technologies were being incorporated for knowledge management. Mahdjoub et al. [45] developed a collaborative design system for mechanical design projects. The system was based on a parametric CAD model, 3D human model simulations, virtual reality tools and a multi-agent system embedded in a product lifecycle management environment. The multi-agent system enabled the localization, classification and extraction of stored knowledge and suggested expert knowledge to designers. A virtual reality platform, data glove and hand tracker enabled engineers to interact in real-time and modify the configuration of the virtual prototype.

Szőke et al. [46] described the utilization of 3D simulations, virtual reality, geographical information systems and an advanced user interface for supporting work in the nuclear industry. They argued that since most data are related to real environments, presenting the data and information to users in the context of their environments enables rapid visual perception and comprehension. The results showed that emerging technologies have huge potential to improve efficiency, safety and transparency in nuclear decommissioning projects.

Sivanathan et al. [47] proposed a virtual-aided design engineering review system (VADER) in which users logged discussions and decisions in real-time using a virtual design environment to promote knowledge capture and facilitate knowledge retrieval. This design engineering review system enabled automatic and unobtrusive logging of multimodal inputs (including audio, video, CAD files, and annotations) and time-phased interactions via an integration and temporal synchronization routine during product design meetings. And the system reformatted the captured content into a readily understandable and accessible format by linking design data with the captured multimodal data from the review meeting.

Roupé et al. [25] proposed a collaborative design system using real-time virtual reality tools to facilitate interactive design work and foster communication. The system enabled different stakeholders to have a first-person perspective and share a common frame of reference for the design environment. They used two collaborative design workshops to evaluate the system in the context of designing new healthcare environments.

Burova et al. [48] described a case study in which asymmetric virtual reality was applied to enable distributed collaboration of geographically dispersed teams working on the development of a series of methods for maintenance and the associated creation of technical documentation. Asymmetry in this study referred to merging the collaboration over two digital platforms: the COVE-VR platform and Microsoft Teams. Based on qualitative evaluation, they demonstrated that distributed asymmetric virtual reality is a low-cost and scalable solution that could be integrated with current industrial practices for remote working and suggested a list of guidelines on how to support the asymmetry between VR and traditional conferencing tools.

### 4.5 Building Information Modelling

There was one primary study which considered how Building Information Modelling supported knowledge management. Building Information Modelling (BIM) is the process of generating and managing data during the building's lifecycle [49]. Ho et al. [50] developed a way to support project managers and jobsite engineers to effectively acquire, share and reuse experience-based knowledge using the Building Information Modelling (BIM) technology within a 3D CAD environment. The proposed BIM-based knowledge sharing system was applied in a case study to demonstrate the effectiveness of sharing knowledge in a construction project.

## 4.6 Cyber-Physical System

Three of the primary studies investigated how cyber-physical systems are being incorporated in knowledge management. A Cyber-Physical System (CPS) is defined as the integration of physical and computational processes or assets in a way that both the virtual and physical parts of the system interact effectively [51].

de Carvalho et al. [52] proposed using sensors with augmented reality technology in a cyber-physical system to record movements, haptic information and other types of implicit knowledge embodied in actions during an industrial set-up. They argued that in a complex industrial domain, augmented reality had considerable potential to improve knowledge sharing by offering new ways that expertise and knowledge embedded in actions could be captured, displayed and shared.

Hoffmann et al. [53] presented an empirical study to investigate the potential of CPS technologies to support and advance knowledge and expertise sharing (KES) practices to support machine set-up in a production environment. Based on a one-year ethnographic study, they developed a knowledge and expertise sharing model then matched CPS-based technical possibilities to the practice-oriented knowledge sharing requirements captured within the model. The model served as a bridge between their ethnographic findings and the design of the CPS. They identified the potential for using CPS to support KES in industrial contexts, which was still at an early stage and relatively unexplored.

Longo et al. [54] developed a human-centric manufacturing paradigm where employees, including both managers and shop-floor operators, were empowered with knowledge about the manufacturing processes. They defined a service-oriented digital twin as "the component that offers an advanced and intuitive knowledge fruition through ubiquitously accessible apps and services." The digital twin worked as a link between the cyber-physical production system (CPPS) and the manufacturing employees. In order to achieve intuitive knowledge retrieval and knowledge creation, a close interaction between the manufacturing employees and the CPPS was facilitated using applications and services provided by the digital twin. These were based on its flexible knowledge structure, augmented reality technologies, and an interactive system.

Dragicevic et al. [55] explored the role of the human-inthe-loop in the Industry 4.0 environment. Industry 4.0 refers to the vision of a fourth industrial revolution. It relates to a new paradigm using smart and intelligent systems, automation, and digitalized production [56]. Dragicevic et al. [55] developed a conceptual model for a smart grid that would provide a holistic view of knowledge-based interactions among human, smart objects, and other computational entities as a mechanism for value creation in Industry 4.0. Smart grid in their paper referred to a power network that integrated the behaviors and actions of all stakeholders connected to it. They argued that, in an Industry 4.0 environment, harnessing computation is to enhance human

### 4.7 Multiple Technologies

There were three primary studies that considered more than one of the main digital technologies. Wang et al. [23] proposed a knowledge management framework based on digital systems for the full lifecycle of a nuclear power plant. The technologies used in the proposed framework included an integrated automatic design and analysis platform, a 3D power plant model, a virtual reality building information model and a big data center. Each technology supported different phases of the work and various stakeholders across the lifecycle of a nuclear powerplant.

Wilkesmann and Wilkesmann [57] developed a theoretical framework to analyze the applications of Industry 4.0 using different digital architectures in various organizational environments. They considered organizations structured in different ways across a spectrum from mechanistic to organic environments. They found that mechanistic environments had a top-down structure in which the digital architecture provided a predetermined hierarchy. Whereas in an organic environment, the digital architecture facilitated innovation through a horizontal structure.

Hannola et al. [58] proposed four applications of digital technologies to empower production workers and facilitate knowledge management in knowledge-intensive production environments. These technologies included augmented reality, social networking platforms, and data mining. However, they only provided conceptual proposals for the application of each technology and no practical applications were described.

### 5 DISCUSSION

This section discusses how the primary studies identified in this systematic review draw on contemporary theories associated with digital technologies and knowledge management and address the top-level research question, i.e., how do digital technologies influence knowledge management in the engineering sector. The use of digital twins with digital frameworks or environments is emerging in a number of industrial sectors to support the design and operation of engineering systems. The prime motivation for this study was to identify the parallel emergence of digital technology in supporting knowledge management associated with these systems which frequently have life cycles that extend beyond the working life of their designers. Hence, the choice of the research question and the positioning of this discussion in knowledge management in the field of engineering. Nevertheless, the discussion attempts to place the outcomes from the systematic review in the broader context of the field of knowledge management by considering the top-level research question using two subsidiary questions. First, from a technical perspective, how digital technologies support the processes of knowledge management in Sub-section 5.1. Second, in sub-section 5.2, how digital technologies influence knowledge management as a social practice, i.e., examining the impact on knowledge, skills and understanding within engineering organizations of digital technology. Existing theories were

found to provide a suitable framework for discussion; hence, no attempt has been made to develop a new theory but instead to synthesize the findings of the primary studies within the current paradigm.

# 5.1 How do digital technologies support knowledge management processes?

One of the most cited knowledge management frameworks was introduced by Alavi and Leidner and has four knowledge processes, namely, knowledge creation, knowledge storage and retrieval, knowledge sharing, and knowledge application [5]. Most of the primary studies considered more than one of these knowledge management processes, and these are discussed below.

### 5.1.1 Knowledge Creation

Knowledge creation comprises ways to facilitate the development of new knowledge. Nonaka and Toyama [59] conceptualized knowledge creation as a dialectical process, in which various contradictions are synthesized through dynamic interactions among individuals, the organization, and the environment. Digitally supported communication can foster knowledge creation by enabling a space for considering multiple viewpoints, for constructing and sharing beliefs, and for allowing expression of new ideas [5].

Kimmerle et al. [31] discussed the development of new knowledge within an organization when people interact with each other using social media platforms. They argued that the exchange between the social system (collective knowledge enabled by shared digital artefacts) and the individual's cognitive system is the basis for the development of new knowledge. Sumbal et al. [36] investigated facilitating knowledge creation through big data. They demonstrated that knowledge was generated through a combination of predictive knowledge obtained from big data analytics with people's tacit knowledge, including their insights and opinions used in decision-making. According to Longo et al. [54], the role of the industrial Internet of Things (IoT) was to enhance and optimize human work by creating new valuable data and information streams instead of fostering a pure machine-to-machine manufacturing environment. Wilkesmann and Wilkesmann [57] discussed what forms of organizational structures and works supported by digital technologies allow the creation of innovations within Industry 4.0. They proposed using the Internet of Things to gain access to new sources of data and information to promote think tanks for new products and ideas. While recently, Ribeiro et al [60] have considered the influence of knowledge management on Industry 4.0 in the context of knowledge creation and work engagement in the telecommunications industry.

### 5.1.2 Knowledge Storage and Retrieval

Knowledge storage and retrieval is related to the management of organizational memory [61]. Organizational memory is defined as the means by which knowledge from the past, experience and events influence present organizational activities [62]. Challenges related to knowledge storage and retrieval have been identified in previous literature; for example, organizations do not know what they know and have weak systems for locating and retrieving knowledge according to Huber [63], and lessons learnt are identified and stored in some systems with different formats, yet they are often left behind and are difficult to find and reuse because of their poor organization [33].

Mahdjoub et al. [45] designed a knowledge management system using a multi-agent system to capture, extract and annotate knowledge during mechanical design projects. They used ontology or a semantic knowledge representation to capture knowledge in a generic way, in order to transition information efficiently and integrate it into an environment for reuse and sharing in support of managing the lifecycle of a product. Similarly, Wan et al. [33] used ontology to represent the semantics of maintenance knowledge in a formal way that a computer could interpret and manage in their proposed advanced content management system. Product-service related knowledge was classified, structured and managed to be shared and reused by different stakeholders.

In contrast to contemporary information and knowledge management systems that are separate from the activity that generates knowledge, Sivanathan et al. [47] proposed a new knowledge capture and reuse para-digm by integrating and automating the knowledge cap-ture process within the natural design review activity. The proposed virtualaided design engineering review (VADER) system enabled the unobtrusive integration of structured and unstructured data via an integration and temporal synchronization routine; and then, systematical-ly organized the captured information for effective retrieval. More recently, de Freitas et al. [64] have considered the benefits and challenges of virtual-reality-based design reviews.

### 5.1.3 Knowledge Sharing

Knowledge sharing occurs when individuals convey knowledge or acquire knowledge from others [65]. The knowledge can be distributed between organizations or within an organization.

One of the motivational aspects of encouraging knowledge sharing is the generation of social capital [66]. Social capital refers to "the sum of the actual and potential resources embedded within, available through, and derived from the network of relationships possessed by an individual or social unit" [67]. Qi and Chau [43] investigated the impact of the usage of enterprise social networking systems on knowledge management processes. Their results aligned with social capital theory by demonstrating that enterprise social networking systems provided a rich source of social capital by increasing interactivity and individuals began, intentionally or unintentionally, to create and share the knowledge they possessed.

Evans et al. [24] proposed a framework for collaboration and knowledge sharing during the product development process through the utilization of social software platforms. The framework conceptualized the potential of social software, and led to recommended guidelines on which technologies might be most appropriate to enhance collaborative and knowledge sharing practices.

It is hard to share knowledge embedded in practice, such as "know how", in a generic form so that anyone might be able to make sense of it [68]. Conventional computer technologies do not meet the needs of sharing knowledge embedded in actions [52]. According to Nona-ka and Takeuchi [3], "know how" does not necessarily need to be transformed into a propositional form for knowledge sharing to take place. A range of other possibilities have been considered by using video, virtual reality, augmented reality and sensor technology. de Carvalho et al. [52] and Hoffmann et al. [53] proposed recording embodied knowledge and aligning it with propositional content via cyber-physical system technology in the context of an industrial set-up. By capturing the knowledge with adequate and timely information in sufficient detail, knowledge embedded in an individual's routines and practices could be easily understood and shared.

The implementation of a Cyber-Physical Production System (CPPS) calls for not only changes on the technical side, such as processes and systems, but also on the side of human-based skills and organizational competencies [16]. The implementation of a CPPS promotes cultural change by empowering workers to become decision makers rather than simply information handlers [69]. Longo et al. [54] developed a manufacturing paradigm where both managers and shop-floor operators were empowered with ubiquitous knowledge about the manufacturing system and processes. "Ubiguitous" in the paper means manufacturing employees can access the knowledge intuitively and guickly everywhere and at any time. A close interaction between the CPPS and the manufacturing employees enabled communication and full information symmetry within and among all the elements of a smart factory.

Information is often related to real environments, presenting the information to people in the context of the relevant environment enables rapid visual perception and comprehension [46]. Ho et al. [50] proposed using building information modelling (BIM) technology within a 3D CAD environment to support engineers in using their experience to update, share and reuse knowledge in the form of graphical representations. They conducted a case study based on construction project and, using questionnaires to evaluate outcomes, found that the BIM-based knowledge management system was effective. Szőke et al. [46] described the use of 3D simulation, virtual reality and an advanced user interface for knowledge management in nuclear decommissioning projects. Their results were obtained from many years of the authors' experience in research and development on the OECD Halden Reactor Project. They demonstrated the potential of virtual plant models and 3D simulation to plan and monitor working progress, to present dynamic situation information, and to train and support communication between stakeholders. Also in the nuclear industry, Wang et al. [23] proposed using a 3D nuclear power plant model to facilitate communication among various stakeholders during the lifecycle of a nuclear plant. This was a conceptual study that proposed a digitally-supported framework for knowledge management framework but its application was not reported.

### 5.1.4 Knowledge Application

Knowledge application refers to the use of knowledge for

a competitive advantage through improvements in the capability of organizations. There are no primary studies focused only on the knowledge application process. One possible explanation is the nature of the knowledge application process, which is aimed at reusing or integrating existing knowledge for future problem-solving or decisionmaking. In the primary studies, the knowledge application process is incorporated with other knowledge processes, but not as the main process being investigated. Instead, it is associated with knowledge storage, which can support access to required knowledge, and with knowledge sharing, which can expand an individual's networks and support collaboration thereby enabling organizational knowledge to be applied across time and space.

### 5.1.5 Summary

The four knowledge processes introduced by Alavi and Leidner [5] have been used as a framework to consider how digital technologies could support knowledge man-agement.

Only five primary studies investigated knowledge creation as their main focus with the most recent being Dragicevic et al. [55]. Hence, this is an area that could benefit from further investigation. Similarly, only five primary studies examined knowledge storage and retrieval as their principal emphasis. All these studies provided detailed information on the preservation of information and knowledge in different forms, while there were fewer details on the performance of knowledge reuse. Since none of these studies mentioned the updating and curation of stored information and knowledge, these topics would appear to an appropriate subject for new studies.

Thirteen primary studies considered knowledge sharing as their main objective. Most of the studies focused on facilitating communication via social media systems. Only two studies mentioned sharing the knowledge and expertise, which are embedded in the actions of individuals, using cyber-physical systems. Incorporating Industry 4.0 technologies in knowledge management is still at an early stage. In order to explore the potential of this type of technology and to identify the issues related to its deployment, more empirical studies are needed across the engineering domain.

Although knowledge management processes are often depicted as sequential in the literature, they are interconnected and interdependent instead of linear [5]. They may never really "start" or "end" and actually run in parallel [70]. If any one of the four processes is weak or fails, the effectiveness and integrity of the overall management of knowledge will suffer [5]. The value of the preserved knowledge depends on its reuse and application, not only recording it [33]. Knowledge creation is strongly related to the knowledge sharing process since social interaction can trigger knowledge creation by enabling individuals to share and develop new ideas [43]. Similar conclusions have been reached recently by González-Tejero et al [71] in the more generic context of small businesses.

To answer the question posed in the title of this section: digital technologies are being to support knowledge management through the coordination and integration of the four knowledge management processes, i.e., knowledge creation, knowledge storage and retrieval, knowledge sharing, and knowledge application. The strongest focus has been on knowledge sharing; however, none of the primary studies considered using digital technologies to combine all four knowledge processes as a holistic activity.

### 5.2 How do digital technologies influence knowledge management as a social practice?

This question is concerned with the everyday practices or the way in which knowledge management is typically and habitually performed and the impact of digital technologies on these processes. Since there are established theories in social science that provide a suitable framework for this discussion, no attempt has been made to develop a new theory. Social science theory recognizes five topics in which digital technologies can facilitate knowledge management as a social practice: repositories, transactive memory systems, common information environments, boundary objects and non-human actors. Although some of these topics were alluded to in several of the primary studies, almost none were explicitly recognized or analyzed by the authors. Each of these topics is discussed in more detail below.

### 5.2.1 Repositories

A repository is a system working to preserve explicit knowledge embedded in documents, reports, lessons learnt, best practices, and procedures. Explicit knowledge is knowledge that can be written down or articulated in language or some other symbolic form [5]. Different types of knowledge management systems have been introduced where knowledge is stored for future retrieval so that preserved knowledge can be reused beyond geographic borders and time limits. Repositories normally preserve explicit knowledge to support the process of knowledge storage and reuse.

Sukumar and Ferrell [35] proposed an enterprise data discovery and recorder tool using big data. It served as a digital record of institutional knowledge and as documentation for the evolution of data elements, queries, and schemas over time. Wan et al. [33] used a content management system to manage explicit knowledge including documents, latest maintenance execution procedures, lessons learnt and best practices in the manufacturing domain. While Sivanathan et al. [47] proposed a design engineering review system which captured time-synchronized multimodal inputs including audio, video, CAD files, drawings, and annotations. Hence, it can be concluded that digital technologies can be used effectively as repositories for knowledge.

### 5.2.2 Transactive Memory Systems

Wegner [72] presented the notion of a transactive memory system that helped group members benefit from distributed knowledge resources. This system developed a group meta-knowledge containing information on who knows what and where the knowledge resides, which enabled individuals to locate the people who possess the needed knowledge [73]. Within a transactive memory system, individuals become sources of external knowledge and rely on each other to pool together complementary knowledge [74].

Enterprise social media not only helps people to solve the specific problem at hand, but also makes locating expertise and connecting knowledge seekers with knowledge contributors more effective [38]. In dispersed organizations with complex structures, it is often difficult to identify and locate individuals who possess highly specific expertise as it can remain hidden and consequently unexploited. Evans et al. [40] proposed using micro-blogging to improve crowdsourcing of knowledge by finding each other, becoming more personally involved, and identifying experts on given subject. Ali et al. [44] developed a model to test the effect of social media on the development of a transactive memory system. According to their results, social media served as an open access meta-knowledge repository by creating individual input footprints and describing who had what type of knowledge.

### 5.2.3 Communication Spaces

Social networking technologies allow knowledge sharing through an online communal knowledge conversation [37]. People feel free to engage in the knowledge conversations rather than paying for the extra cost of codifying knowledge for input to a formal repository [75]. An individual's knowledge network is extended, with the help of social networking platforms, to reach beyond the beyond the formal communication networks and encourage people to naturally interact with one another in their daily work [43]. Based on primary studies, social networking technologies have great potential to facilitate communication by supporting rapid knowledge flow between people working across geographical boundaries [76]; breaking down knowledge silos in organizations [40]; facilitating collaboration between multi-functional business units and multitier supply chains [24]; and identifying the right people with whom to communicate as part of the process of engineering design [39].

### 5.2.4 Boundary Objects

Star and Griesemer [77] described boundary objects as "...objects which both inhabit several interesting social worlds and satisfy the information requirements of each of them. They are plastic enough to adapt to local needs and the constraints of the several parties employing them, yet robust enough to maintain a common identity across sites." Boundary objects are critical to knowledge sharing. They serve as translation mechanisms for different ideas, view-points and perceptions across otherwise difficult to traverse social boundaries [78] and enable collaboration between different communities of practice or social worlds [79]. However, successful collaboration requires the communities or worlds to have a common understanding and shared interest in the boundary object [80].

The virtual-aided design engineering review (VADER) system proposed by Sivanathan et al. [47] functioned as a boundary object for users involved all phases of the lifecycle of a product while it maintained a common identity as design review system. Distributed users could access, share and reuse knowledge for various tasks via its collaborative 3D interface and auxiliary web interface. Similarly, the framework for knowledge management proposed by Wang et al. [23] for the nuclear industry can also be viewed as a boundary object because it was designed to provide a range of services to different users, including digital R&D, design, verification and validation, manufacture, construction, operation and management.

#### 5.2.5 Non-Human Actors

Actors can be defined as entities that perform knowledgebased activities [55]. Digital technology working as a nonhuman actor is not merely a passive container for transmitting knowledge, rather it is an actor interacting with people and the environment in doing knowledge work [80]. Sensors, machines and equipment are equipped with embedded local intelligence, which makes them smart objects interwoven over the cloud in order to cooperate with each other [81]. Dragicevic et al. [55] proposed a theory-based knowledge dynamics model in a smart grid scenario that provided a holistic view on knowledge-based interactions among human, smart objects and other actors as a mechanism of value co-creation in Industry 4.0. The deployment of Industry 4.0 had not only a technical but also a human dimension in the interaction of people with technologies [82].

### 5.2.6 Summary

In summary, to address the question at the start of this section concerning how digital technologies influence knowledge management as a social practice, five roles played by digital technology have been identified, namely as repositories, transactive memory systems, communication spaces, boundary objects and non-human actors.

However, people, processes, and technology need to be incorporated together to support knowledge management [10] because digital technologies acting alone provide little value to an organization [83] since employee competency is crucial in enabling an organization's transformation toward digitalization [84]. Rather than a purely technologydriven approach, there is an evident need for research on people-centered design to properly understand the requirements in social-technical contexts, where such systems or tools for knowledge management will be deployed [52]. The developed tools must be simple to use and embedded into the current systems that people are using, and not developed into "yet another tool" that becomes redundant over a period of time [40], especially given that the costs of digital technologies appear in the immediate or short-term whereas the benefits of knowledge management may only become apparent in the long-term.

### 6 CONCLUSIONS

This paper presents a systematic literature review on how digital environments influence knowledge management in the engineering sector. The following conclusions can be drawn from the analysis presented:

 3097 results were initially identified, and these were then selected systematically leaving a total of 27 primary studies. These primary studies were then reviewed in more detail (Section 4 and Section 5).

- This analysis has revealed that digital technologies have been employed to support knowledge management in the engineering sector. For the purposes of this study knowledge management was broken down into four sub-processes, i.e., knowledge creation, knowledge storage and retrieval, knowledge sharing, and knowledge application. Among the primary studies, the knowledge sharing process was most prevalent, featuring in (48%) of them, while none of them had knowledge application as their main focus. Furthermore, none of the primary studies investigated knowledge management as a holistic process but instead focused on one or two knowledge processes.
- Almost no studies have investigated the social implications of the adoption of digital technologies in engineering knowledge management. Five people-centric roles played by digital technologies in knowledge management were identified from the primary studies, namely as repositories, transactive memory systems, communication spaces, boundary objects, and non-human actors. It was evident that the ability of digital technologies to perform all of these roles was not considered explicitly within the primary studies and this would therefore be an appropriate area for further investigation.

The top-level question posed at the start of this review was: 'how do digital technologies influence knowledge management in the engineering sector?' The answer can be summarised as knowledge management in the engineering sector is being somewhat influenced by digital technologies but in a limited manner due to piecemeal implementations rather than holistic approaches. As a consequence, there is little or no effect on social practices, i.e., changes in the way knowledge management is habitually performed are not apparent. Therefore, there are significant opportunities for engineering organizations to enhance their competitive advantage and organizational value by improving their speed and efficiency in creating and sharing knowledge through the effective deployment of digital technologies.

### ACKNOWLEDGEMENTS

This research is jointly funded by the EPSRC Centre for Doctoral Training in NGN (Next Generation Nuclear) and the University of Manchester.

### REFERENCES

- J.S. Edwards, "Knowledge management in the energy sector: review and future directions," *International Journal of Energy Sector Management*, vol. 2, no. 2, pp. 197-217, Jun. 2008.
- [2] B. Kogut and U. Zander, "Knowledge of the firm, combinative capabilities, and the replication of technology," *Organization Science*, vol. 3, no. 3, pp. 383-397, Aug. 1992.
- [3] Nonaka, I. o Nonaka, N. Ikujiro and H. Takeuchi, *The Knowledge-Creating Company: How Japanese Companies Create the Dynamics of Innovation*, New York, USA: Oxford University Press, 1995.
- [4] R.M. Grant, "Toward a knowledge-based theory of the firm," *Strategic Management Journal*, vol. 17, no. S2, pp. 109-122, Dec. 1996.

- [5] M. Alavi and D.E. Leidner, "Knowledge management and knowledge management systems: Conceptual foundations and research issues," *MIS Quarterly*, vol. 25, no. 1, pp. 107-136, Mar. 2001.
- [6] E. Glaessgen, D. Stargel, "The digital twin paradigm for future NASA and US Air Force vehicles," *Proc 53rd AIAA/ASME/ASCE/AHS/ASC Structures, Structural Dynamics and Materials Conference*, AIAA paper 2012-2018, Apr. 2012.
- [7] E.A. Patterson, R.J. Taylor, M. Bankhead, "A framework for an integrated nuclear digital environment," *Progress in Nuclear Energy*, vol. 87, pp. 97-103, Mar. 2016.
- [8] T. Sarina, "Enhancing knowledge management (KM) in the fourth industrial revolution era: The role of human resource systems," in *The Palgrave Handbook of Knowledge Management*, J. Syed, P. Murray, D. Hislop, Y. Mouzughi, Eds., Cham, Switzerland: Palgrave Macmillan, 2018, pp. 411-435.
- [9] M.M. Bonanomi, D.M. Hall, S. Staub-French, A. Tucker and C.M.L. Talamo, "The impact of digital transformation on formal and informal organizational structures of large architecture and engineering firms," *Engineering, Construction and Architectural Management*, vol. 27, no. 4, pp. 872-892, Nov. 2019.
- [10] R. Ruggles, "The state of the notion: knowledge management in practice," *California Management Review*, vol. 40, no. 3, pp. 80-89, Apr. 1998.
- [11] G.D. Bhatt, "Knowledge management in organizations: examining the interaction between technologies, techniques, and people," *Journal of Knowledge Management*, vol. 5, no. 1, pp. 68-75, Mar. 2001.
- [12] Davies, H. Russell, C. Murphy, M. Alhefdhi, R. Olayinka, S. Traynor, R.R. Pushpa, C.H. Clark, C. White-Robinson and V. Ribiere, "The changing role of the knowledge manager", *Tomorrow's KM: Innovation, Best Practice, and the Future of Knowledge Management*, UK: ARK Group, 2019, pp. 23-31.
- [13] S. Nidhra, M. Yanamadala, W. Afzal and R. Torkar, "Knowledge transfer challenges and mitigation strategies in global software development—A systematic literature review and industrial validation," *International Journal* of Information Management, vol. 33, no. 2, pp. 333-355, Apr. 2013.
- [14] Y.A. Ahmed, M.N. Ahmad, N. Ahmad and N.H. Zakaria, "Social media for knowledge-sharing: A systematic literature review," *Telematics and Informatics*, vol. 37, pp. 72-112, Apr. 2019.
- [15] H. Inkinen, "Review of empirical research on knowledge management practices and firm performance," *Journal of Knowledge Management*, vol. 20, no. 2, pp. 230-257, Apr. 2016.
- [16] M.F. Manesh, M.M. Pellegrini, G. Marzi and M. Dabic, "Knowledge management in the fourth industrial revolution: Mapping the literature and scoping future avenues," *IEEE Transactions on Engineering Management*, vol. 68, no. 1, pp. 289-300, Jan. 2020.
- [17] J. Child and L.H. Hsieh, "Decision mode, information and network attachment in the internationalization of SMEs: A configurational and contingency analysis," *Journal of World Business*, vol. 49, no. 4, pp. 598-610, Oct. 2014.
- [18] T.H. Davenport and L. Prusak, *Information Ecology: Mastering the Information and Knowledge Environment*, New York, USA: Oxford University Press, 1997.
- [19] L Fahey and L. Prusak, "The eleven deadliest sins of knowledge management," *California Management Review*, vol. 40, no. 3, pp. 265-276, Apr. 1998.
- [20] J.A. Simpson and E.S.C. Weiner, "Oxford English dictionary," Oxford, U.K.: Oxford University Press, 1989.
- [21] G. Madhavan, *Think Like an Engineer: Inside the Minds That are Changing Our Lives*, London, U.K.: Oneworld Publications, 2015.

- [22] A.A. Mishra and R. Shah, "In union lies strength: Collaborative competence in new product development and its performance effects," *Journal* of Operations Management, vol. 27, no. 4, pp. 324-338, Aug. 2009.
- [23] M. Wang, M. Zheng, L. Tian, Z. Qiu and X. Li, "A full life cycle nuclear knowledge management framework based on digital system," *Annals of Nuclear Energy*, vol. 108, pp. 386-393, Oct. 2017.
- [24] R.D. Evans, J.X. Gao, N. Martin and C. Simmonds, "A new paradigm for virtual knowledge sharing in product development based on emergent social software platforms," *Proceedings of the Institution of Mechanical Engineers, Part B: Journal of Engineering Manufacture*, vol. 232, no. 13, pp. 2297-2308, Nov. 2018.
- [25] M. Roupé, M. Johansson, L. Maftei, R. Lundstedt and M. Viklund-Tallgren, "Virtual Collaborative Design Environment: Supporting Seamless Integration of Multitouch Table and Immersive VR," *Journal of Construction Management and Engineering*, vol. 146, no. 12, Dec. 2020, Art no. 04020132.
- [26] B.A. Kitchenham and S. Charters, "Guidelines for performing systematic literature reviews in software engineering," Keele University and Durham University Joint Report, EBSE 2007-001, Jul. 2007.
- [27] M.K. Linnenluecke, M. Marrone, A.K. Singh, "Conducting systematic literature reviews and bibliometric analyses," *Australian Journal of Management*, vol. 45, no. 2, pp. 175-194, May. 2020.
- [28] D. Tranfield, D. Denyer and P. Smart, "Towards a methodology for developing evidence-informed management knowledge by means of systematic review," *British Journal of Management*, vol. 14, no. 3, pp. 207-222, Sep. 2003.
- [29] A.D. Oxman, "Systematic reviews: checklists for review articles," *BMJ*, vol. 309, no. 6955, pp. 648-651, Sep. 1994.
- [30] R.A.B. Ouriques, K. Wnuk, T. Gorschek and R.B. Svensson, "Knowledge management strategies and processes in agile software development: a systematic literature review," *International Journal of Software Engineering and Knowledge Engineering*, vol. 29, no. 03, pp. 345-380, Mar. 2019.
- [31] J. Kimmerle, U. Cress and C. Held, "The interplay between individual and collective knowledge: Technologies for organizational learning and knowledge building," *Knowledge Management Research & Practice*, vol. 8, no. 1, pp. 33-44, Mar. 2010.
- [32] P. Kirsch, A. Hine and T. Maybury, "A model for the implementation of industry-wide knowledge sharing to improve risk management practice," *Safety Science*, vol. 80, pp. 66-76, Dec. 2015.
- [33] S. Wan, D. Li, J. Gao, R. Roy and F. He, "A collaborative machine tool maintenance planning system based on content management technologies," *The International Journal of Advanced Manufacturing Technology*, vol. 94, no. 5, pp. 1639-1653, Feb. 2018.
- [34] Y. Abbas, A. Martinetti, M. Rajabalinejad, F. Schuberth and L. van Dongen, "Facilitating digital collaboration through knowledge management: a case study," *Knowledge Management Research & Practice*, Feb. 2022, doi: 10.1080/14778238.2022.2029597
- [35] S.R. Sukumar and R.K. Ferrell, "'Big Data' collaboration: Exploring, recording and sharing enterprise knowledge," *Information Services & Use*, vol. 33, no. 3-4, pp. 257-270, Jan. 2013.
- [36] M.S. Sumbal, E. Tsui and E.W. See-to, "Interrelationship between big data and knowledge management: an exploratory study in the oil and gas sector," *Journal of Knowledge Management*, vol. 21, no. 1, pp. 180-196, Feb. 2017.
- [37] Majchrzak, S. Faraj, G.C. Kane and B. Azad, "The contradictory influence of social media affordances on online communal knowledge sharing," *Journal of Computer-Mediated Communication*, vol. 19, no. 1, pp. 38-55, Oct. 2013.
- [38] R. Beck, I. Pahlke and C. Seebach, "Knowledge exchange and symbolic action in social media-enabled electronic networks of practice," *MIS Quarterly*, vol. 38, no. 4, pp. 1245-1270, Dec. 2014.

- [39] J.A. Gopsill, H.C. McAlpine and B.J. Hicks, "Supporting engineering design communication using a custom-built social media tool - PartBook," *Advanced Engineering Informatics*, vol. 29, no. 3, pp. 523-548, Aug. 2015.
- [40] R.D. Evans, J.X. Gao, S. Mahdikhah, M. Messaadia and D. Baudry, "Crowdsourcing user-contributed solutions to aerospace product development issues through micro-blogging," *Electronic Journal of Knowledge Management*, vol. 14, no. 2, pp. 126-136, Jun. 2016.
- [41] V. Scuotto, M. Del Giudice and E.G. Carayannis, "The effect of social networking sites and absorptive capacity on SMES' innovation performance," *The Journal of Technology Transfer*, vol. 42, no. 2, pp. 409-424, Apr. 2017.
- [42] W. Van Osch and C.W. Steinfield, "Strategic visibility in enterprise social media: Implications for network formation and boundary spanning," *Journal of Management Information Systems*, vol. 35, no. 2, pp. 647-682, Apr. 2018.
- [43] C. Qi and P.Y.K. Chau, "Will enterprise social networking systems promote knowledge management and organizational learning? An empirical study," *Journal of Organizational Computing and Electronic Commerce*, vol. 28, no. 1, pp. 31-57, Jan. 2018.
- [44] Ali, W. Bahadur, N. Wang, A. Luqman and A.N. Khan, "Improving team innovation performance: Role of social media and team knowledge management capabilities," *Technology in Society*, vol. 61, May. 2020, Art no. 101259.
- [45] M. Mahdjoub, D. Monticolo, S. Gomes and J.C. Sagot, "A collaborative design for usability approach supported by virtual reality and a multi-agent system embedded in a PLM environment," *Computer-Aided Design*, vol. 42, no. 5, pp. 402-413, May. 2010.
- [46] Szőke, M.N. Louka, T.R. Bryntesen, S.T. Edvardsen and J. Bratteli, "Comprehensive support for nuclear decommissioning based on 3D simulation and advanced user interface technologies," *Journal of Nuclear Science and Technology*, vol. 52, no. 3, pp. 371-387, Mar. 2015.
- [47] Sivanathan, J.M. Ritchie and T. Lim, "A novel design engineering review system with searchable content: knowledge engineering via real-time multimodal recording," *Journal of Engineering Design*, vol. 28, no. 10-12, pp. 681-708, Dec. 2017.
- [48] Burova, P.B. Palma, P. Truong, J. Mäkelä, H. Heinonen, J. Hakulinen, K. Ronkainen, R. Raisamo, M. Turunen and S. Siltanen, "Distributed asymmetric virtual reality in industrial context: Enhancing the collaboration of geographically dispersed teams in the pipeline of maintenance method development and technical documentation creation," *Applied Sciences*, vol. 12, no. 8, Apr. 2022, Art no. 3728.
- [49] E. Epstein, Implementing Successful Building Information Modeling, Norwood, USA: Artech House, 2012.
- [50] S.P. Ho, H.P. Tserng and S.H. Jan, "Enhancing knowledge sharing management using BIM technology in construction," *The Scientific World Journal*, vol. 2013, 2013, Art no. 170498.
- [51] E.A. Lee and S.A. Seshia, Introduction to Embedded Systems: A Cyber-Physical Systems Approach, Cambridge, MA, USA: MIT Press, 2016.
- [52] A.F.P. de Carvalho, S. Hoffmann, D. Abele, M. Schweitzer, P. Tolmie, D. Randall and V. Wulf, "Of embodied action and sensors: Knowledge and expertise sharing in industrial set-up," *Computer Supported Cooperative Work (CSCW)*, vol. 27, no. 3, pp. 875-916, Dec. 2018.
- [53] S. Hoffmann, A.F.P. de Carvalho, D. Abele, M. Schweitzer, P. Tolmie and V. Wulf, "Cyber-physical systems for knowledge and expertise sharing in manufacturing contexts: Towards a model enabling design," *Computer Supported Cooperative Work (CSCW)*, vol. 28, no. 3, pp. 469-509, Jun. 2019.
- [54] F. Longo, L Nicoletti and A. Padovano, "Ubiquitous knowledge empowers the Smart Factory: The impacts of a Service-oriented Digital Twin on enterprises' performance," *Annual Reviews in Control*, vol. 47, pp. 221-236, Jan. 2019.

- [55] N. Dragicevic, A. Ullrich, E. Tsui and N. Gronau, "A conceptual model of knowledge dynamics in the industry 4.0 smart grid scenario," *Knowledge Management Research & Practice*, vol. 18, no. 2, pp. 199-213, 2020.
- [56] P.K. Muhuri, A.K. Shukla and A. Abraham, "Industry 4.0: A bibliometric analysis and detailed overview," *Engineering Applications of Artificial Intelligence*, vol. 78, pp. 218-235, Feb. 2019.
- [57] M. Wilkesmann and U. Wilkesmann, "Industry 4.0 organizing routines or innovations?," VINE Journal of Information and Knowledge Management Systems, vol. 48, no. 2, pp. 238-254, May. 2018.
- [58] L. Hannola, A. Richter, S. Richter and A. Stocker, "Empowering production workers with digitally facilitated knowledge processes–a conceptual framework," *International Journal of Production Research*, vol. 56, no. 14, pp. 4729-4743, Jul. 2018.
- [59] Nonaka and R. Toyama, "A firm as a dialectical being: towards a dynamic theory of a firm," *Industrial and Corporate Change*, vol. 11, no. 5, pp. 995-1009, Nov. 2002.
- [60] V.B. Ribeiro, D. Nakano, Jr, J. Muniz, R.B.D. Oliveira, "Knowledge management and Industry 4.0: a critical analysis and future agenda," *Gestão & Produção*, prod. 29, Dec. 2022.
- [61] J.P. Walsh and G.R. Ungson, "Organizational memory," *The Academy of Management Review*, vol. 16, no. 1, pp. 57-91, Jan. 1991.
- [62] E.W. Stein and V. Zwass, "Actualizing organizational memory with information systems," *Information Systems Research*, vol. 6, no. 2, pp. 85-117, Jun. 1995.
- [63] G.P. Huber, "Organizational learning: The contributing processes and the literatures," *Organization Science*, vol. 2, no. 1, pp. 88-115, Feb. 1991.
- [64] F.V. de Freitas, M.V.M. Gomes, I. Winkler, "Benefits and challenges of virtual-reality-based industrial usability testing and design reviews: A patents landscape and literature review," *Applied Sciences*, vol. 12, no. 3, p. 1755, Feb. 2022.
- [65] C.J. Chen and S.W. Hung, "To give or to receive? Factors influencing members' knowledge sharing and community promotion in professional virtual communities," *Information & Management*, vol. 47, no. 4, pp. 226-236, May. 2010.
- [66] Y.S. Hau, B. Kim, H. Lee and Y.-G. Kim, "The effects of individual motivations and social capital on employees' tacit and explicit knowledge sharing intentions," *International Journal of Information Management*, vol. 33, no. 2, pp. 356-366, Apr. 2013.
- [67] J. Nahapiet and S. Ghoshal, "Social capital, intellectual capital, and the organizational advantage," *Academy of Management Review*, vol. 23, no. 2, pp. 242-266, Apr. 1998.
- [68] M.S. Ackerman, J. Dachtera, V. Pipek and V. Wulf, "Sharing knowledge and expertise: The CSCW view of knowledge management," *Computer Supported Cooperative Work (CSCW)*, vol. 22, no. 4, pp. 531-573, Aug. 2013.
- [69] F. Ansari and U. Seidenberg, "A Portfolio for Optimal Collaboration of Human and Cyber Physical Production Systems in Problem-Solving," in 13<sup>th</sup> International Conference on Cognition and Exploratory Learning in Digital Age, Mannheim, Germany, Oct. 2016, pp. 311-314.
- [70] K. Wiig, *People-focused knowledge management*, Burlington, MA, USA: Elsevier Butterworth-Heinemann, 2004.
- [71] C.B. González-Tejero, C.M. Molina, "Training, corporate culture and organizational work models for the development of corporate entrepreneurship in SMEs," *Journal of Enterprising Communities: People and Places in the Global Economy*, Feb. 2022.
- [72] D.M. Wegner, "Transactive memory: A contemporary analysis of the group mind," in *Theories of group behavior: Springer Series in Social Psychology*, B. Mullen, G.R. Goethals, Eds., New York, NY, USA: Springer, 1987, pp. 185-208.

- [73] L Argote and Y. Ren, "Transactive memory systems: A micro foundation of dynamic capabilities," *Journal of Management Studies*, vol. 49, no. 8, pp. 1375-1382, Dec. 2012.
- [74] L. Argote, Organizational Learning: Creating, Retaining and Transferring Knowledge, 2<sup>rd</sup> ed. New York, NY, USA: Springer, 2013.
- [75] J. Fulk and Y.C. Yuan, "Location, motivation, and social capitalization via enterprise social networking," *Journal of Computer-Mediated Communication*, vol. 19, no. 1, pp. 20-37, Oct. 2013.
- [76] S. Panahi, J. Watson and H. Partridge, "Conceptualizing social media support for tacit knowledge sharing: physicians' perspectives and experiences," *Journal of Knowledge Management*, vol. 20, no. 2, pp. 344-363, Apr. 2016.
- [77] S.L. Star and J.R. Griesemer, "Institutional ecology, translations' and boundary objects: Amateurs and professionals in Berkeley's Museum of Vertebrate Zoology, 1907-39," *Social Studies of Science*, vol. 19, no. 3, pp. 387-420, Aug. 1989.
- [78] Doolin and L. McLeod, "Sociomateriality and boundary objects in information systems development," *European Journal of Information Systems*, vol. 21, no. 5, pp. 570-586, Sep. 2012.
- [79] E. Wenger, "Communities of practice: Learning as a social system," Systems Thinker, vol. 9, no. 5, pp. 2-3, Jun. 1998.
- [80] S. Newell, "Managing knowledge and managing knowledge work: what we know and what the future holds," *Journal of Information Technology*, vol. 30, no. 1, pp. 1-17, Mar. 2015.
- [81] S. Wang, J. Wan, D. Li and C. Zhang, "Implementing smart factory of industry 4.0: an outlook," *International Journal of Distributed Sensor Networks*, vol. 12, no. 1, Jan. 2016, Art no. 3159805.
- [82] S.J. Oks, A. Fritzsche and K.M. Möslein, "An application map for industrial cyber-physical systems," in *Industrial Internet of Things, Springer Series in Wireless Technology*, S. Jeschke, C. Brecher, H. Song, D. Rawat, Eds., Cham, Switzerland: Springer, 2017, pp. 21-46.
- [83] G. Vial, "Understanding digital transformation: A review and a research agenda," *The Journal of Strategic Information Systems*, vol. 28, no. 2, pp. 118-144, Jun. 2019.
- [84] C. Blanka, B. Krumay, D. Rueckel, "The interplay of digital transformation and employee competency: A design science approach," *Technological Forecasting and Social Change*, vol. 178, p. 121575, May. 2022.



**Yuxin Yao** is a PhD candidate in the School of Engineering at the University of Manchester. She received her MSc degree in mechanical engineering in the University of Manchester. Her current research interests include knowledge management in engineering sectors, social network analysis.



**EANN A. PATTERSON** received a B. Eng. in Mechanical Engineering in 1982 and a PhD in Engineering in 1985, both from the University of Sheffield, UK. He has been the AA Griffith Chair of Structural Materials and Mechanics at the University of Liverpool since 2011 and was appointed Dean of the School of Engineering in 2019. Previously he was Chair of the Department of Mechanical Engineering

at Michigan State University and Head of the Department of Mechanical Engineering at the University of Sheffield. His research interests are in experimental mechanics and application of measurement data to support computational modelling in the aerospace, biomedical and energy industries. Prof. Patterson was awarded a Royal Society Wolfson Research Merit Award in 2011 and is a chartered engineer, a Fellow of both the Royal Academy of Engineering (FREng), and the Institution of Mechanical Engineers (FIMechE).



**Richard Taylor** is BNFL Chair in Nuclear Energy Systems and Associate Director of the Dalton Nuclear Institute. Before joining Manchester, he was the National Nuclear Laboratory's (NNL's) Chief Engineer where he carried accountability for engineering input to around £80 million of technical projects per annum. As Head of Technology at NNL he led multi-million pound technical programmes including

research underpinning clean-up operations at Sellafield, options for the disposition of special nuclear material in the UK and the feasibility of longer term spent nuclear fuel surface storage.